\begin{table}%t2
%\centering
\par
  \caption{25 rediscovered DQ white dwarfs in the same format as
    Table~\ref{tabledq}. 20 of these are analyzed also in
    DB05. Information on the remaining 5 can be found in the
    \cite{McCook.Sion99} catalog (WD~1426+613),
    \cite{Liebert.Harris.ea03}, \cite{Carollo.Bucciarelli.ea06}, and
    \cite{Harris.Liebert.ea03}.\label{tabledq1} }
      \begin{tabular}{lrrrcccc}
        \hline\hline
	\noalign{\smallskip}
        SDSS name&MJD&Plate&Fid&$ \Teff$ [K]& [C/He] &$ \sigma$
        [C/He]&Spectral type/Ref\\ 
	\noalign{\smallskip}
        \hline
        SDSS~J000011.57$-$085008.4&52~143&650&450&8042&--5.46&0.08&\\ 
        SDSS~J000807.54$-$103405.6&52~141&651&199&7768&--5.66&0.06&\\ 
        SDSS~J002531.50$-$110800.9&52~145&653&86&8367&--4.96&0.01&\\ 
        SDSS~J015433.57$-$004047.2&51~871&403&268&7435&--5.89&0.03&\\ 
        SDSS~J015441.75+140308.0&51~877&430&558&6511&--6.89&0.01&\\ 
        SDSS~J032054.11$-$071625.4&51~924&460&236&6266&--5.45&\\ 
        SDSS~J033218.22$-$003722.1&51~810&415&240&8600&--4.62&0.03&\\ 
        SDSS~J090157.92+575135.9&51~924&483&600&&&&
                 	\cite{Liebert.Harris.ea03} \\ 
        SDSS~J091922.18+023605.0&51~929&473&458&11566&&&\\ 
        SDSS~J093537.00+002422.0&52~314&476&461&4958&--6.19&0.02&
                	\cite{Harris.Liebert.ea03} \\ 
        SDSS~J094004.64+021022.6&52~026&477&493&7283&--5.95&0.01&\\ 
        SDSS~J095137.60+624348.7&51~943&487&227&8388&--5.11&0.10&\\ 
        SDSS~J113359.94+633113.2&52~059&597&139&12082&&&\\ 
        SDSS~J114851.68$-$012612.8&52~056&329&578&9174&--3.73&0.01&\\ 
        SDSS~J125359.61+013925.6&52~026&523&252&8282&--4.98&0.02&\\ 
        SDSS~J123752.12+415625.8&53~090&1454&146&5846&--5.51&0.01&
	                \cite{Carollo.Bucciarelli.ea06} \\ 
        SDSS~J132858.20+590851.0&52~411&959&504&&&&
	                \cite{Liebert.Harris.ea03} \\ 
        SDSS~J142728.30+611026.4&52~368&607&379&6427&--6.83&0.01&WD~1426+613  \\ 
        SDSS~J144407.25+043446.8&52~026&587&418&9449&--3.65&0.05&\\ 
        SDSS~J144808.07$-$004755.9&51~662&308&145&7063&--6.50&0.04&\\ 
        SDSS~J154810.66+562647.7&52~072&617&551&8119&--5.46&0.07&\\ 
        SDSS~J155413.53+033634.5&52~023&595&373&6512&--6.94&0.03&\\ 
        SDSS~J164328.54+400204.3&52~050&630&386&7144&--6.20&0.07&\\ 
        SDSS~J165538.51+372247.1&52~071&632&92&8997&--4.75&0.06&\\ 
        SDSS~J205316.34$-$070204.3&52~176&636&267&6382&--5.45&0.02&\\ 
	\noalign{\smallskip}
        \hline
      \end{tabular}
\end{table}